Author's response to reviews

Title: Quality of life measurements as an indicator for timing of support after oesophagectomy for cancer: a prospective study

Authors:

Marlene Malmström (marlene.malmstrom@med.lu.se)
Rosemarie Klefsgard (rosmarie.klefsgard@skane.se)
Bodil Ivarsson (bodil.ivarsson@med.lu.se)
Maria Roman (maria.roman@skane.se)
Jan Johansson (jan.johansson@med.lu.se)

Version: 3 Date: 22 December 2014

Author's response to reviews: see over
Dear Editor and reviewers,

Thank you very much for the review of the submitted paper, MS: 1538497230101554 - *Quality of Life measurements as an indicator for timing of support after oesophagectomy for cancer- A prospective study*. We have now made changes in the manuscript according to the constructive criticism made by the reviewer as well as the editorial requests.

Kind regards
Marlene Malmström and co authors
2014-12-22

---

**Reviewer: Krzysztof Tomaszewski**

*The current manuscript is a considerable improvement to the previous version. However I still think there is room for improvement - especially in terms of style in the newly added parts.*

- The newly added part has now been revised in terms of style to better fit the manuscript.

*Needs some language corrections before being published.*

- The manuscript has now been through professional language editing.

---

**Reviewer: David J Bowrey**

*There are still a number of spelling errors in the Figures which should be corrected.*

- The manuscript has now been through professional language editing.